Author’s response to reviews

Title: Emotional stimuli candidates for behavioural intervention in the prevention of early childhood caries: a pilot study

Authors:

Michaela Bartosova (michaela.bartosova@fnusa.cz)

Miroslav Svetlak (miroslav.svetlak@fnbrno.cz)

Martina Kukletova (makukl@med.muni.cz)

Petra Borilova Linhartova (plinhart@med.muni.cz)

Lydie Izakovicova Holla (holla@med.muni.cz)

Version: 2 Date: 16 Aug 2018

Author’s response to reviews:

BMC Oral Health

Editor

Liam Messin

Brno, August 16, 2018


authors: Michaela Bartosova, Miroslav Svetlak, Martina Kukletova, Petra Borilova Linhartova, Lydie Izakovicova Holla

Dear Editor,

We corrected our manuscript according to your suggestions from August 16, 2018.
• Based on the editor’s request, in the consent for publication section, we state “written consent for publication was obtained from patients’ caregivers in this manuscript.” (Wording of the sentence was slightly modified due to the age of our patients - 1.19 +/- 0.32 years).
• We deleted the statement “and we have the consent for publication from the people/guardians in the pictures” from the ethics approval and consent to participate section as suggested.

Yours sincerely,

Lydie Izakovicova Holla, MD., Ph.D.
Professor
Clinic of Stomatology
and Department of Pathophysiology
Medical Faculty, Masaryk University
Pekarska 53, 656 91 Brno
Czech Republic
Fax: +420-543183406
e-mail: holla@med.muni.cz